#### AGU100 ADVANCING EARTH AND SPACE SCIENCE

ŗ

# **Geophysical Research Letters**

# **RESEARCH LETTER**

10.1029/2019GL083770

#### **Key Points:**

- Interannual variability of peak snowpack amount in the western United States will decrease in regions transitioning from snow to rain
- Peak snowpack timing will become more interannually variable and occur across a broader range of months
- Consecutive snow droughts will become much more common across mountains of the western United States

**Supporting Information:** 

Supporting Information S1

# Correspondence to:

A. M. Marshall, adriennemarshall@uidaho.edu

#### Citation:

Marshall, A. M., Abatzoglou, J. T., Link, T. E., & Tennant, C. J. (2019). Projected changes in interannual variability of peak snowpack amount and timing in the Western United States. *Geophysical Research Letters*, 46, 8882–8892. https:// doi.org/10.1029/2019GL083770

Received 17 MAY 2019 Accepted 10 JUL 2019 Accepted article online 8 AUG 2019 Published online 9 AUG 2019

#### ©2019. The Authors.

This is an open access article under the terms of the Creative Commons Attribution-NonCommercial-NoDerivs License, which permits use and distribution in any medium, provided the original work is properly cited, the use is non-commercial and no modifications or adaptations are made.

# Projected Changes in Interannual Variability of Peak Snowpack Amount and Timing in the Western United States

Adrienne M. Marshall<sup>1</sup>, John T. Abatzoglou<sup>2</sup>, Timothy E. Link<sup>3</sup>, and Christopher J. Tennant<sup>4</sup>

<sup>1</sup>Water Resources Program, University of Idaho, Moscow, ID, USA, <sup>2</sup>Department of Geography, University of Idaho, Moscow, ID, USA, <sup>3</sup>Department of Forest, Rangeland, and Fire Sciences, University of Idaho, Moscow, ID, USA, <sup>4</sup>Department of Geography, University of California, Berkeley, Berkeley, CA, USA

**Abstract** Interannual variability of mountain snowpack has important consequences for ecological and socioeconomic systems, yet changes in variability have not been widely examined under future climates. Physically based snowpack simulations for historical (1970–1999) and high-emission scenario (RCP 8.5) mid-21st century (2050–2079) periods were used to assess changes in the variability of annual maximum snow water equivalent (SWE<sub>max</sub>) and SWE<sub>max</sub> timing across the western United States. Models show robust declines in the interannual variability of SWE<sub>max</sub> in regions where precipitation is projected to increase frequency of consecutive snow drought years (SWE<sub>max</sub> < historical 25th percentile) is projected to increase from 6.6% to 42.2% of years. Models also project increases in the variability of SWE<sub>max</sub> timing, suggesting reduced reliability of when SWE<sub>max</sub> occurs. Differences in physiography and regional climate create distinct spatial patterns of change in snowpack variability that will require adaptive strategies for environmental resource management.

**Plain Language Summary** A wealth of research has established that warming temperatures associated with climate change in the western United States will generally reduce snowpack accumulation and result in earlier snowmelt timing, with important consequences for water resources and ecosystems. However, changes in the variability of snowpack conditions between years have not been well established. We analyze simulated snowpack data for historical and future climate scenarios and find that changes in variability differ across the western United States. Variability of annual maximum snowpack between years decreases while the timing of peak snow accumulation becomes more variable, particularly in areas transitioning from snow- to rain-dominated precipitation. We also find that consecutive years with very low or early snowpack will become much more frequent. These findings highlight the need to consider changes in snowpack variability in climate change impact assessments and adaptation planning.

# 1. Introduction

Anthropogenic climate change is altering water resources in the western United States, with decreasing mountain snowpack across the region (Mote et al., 2018; Pierce et al., 2008) and earlier runoff timing in basins that supply water to humans and ecosystems (Barnett et al., 2008). The effects of warming on changes in average snowpack conditions are well characterized for historical (Hamlet & Lettenmaier, 2007; Knowles et al., 2006; Mote, 2006; Pierce et al., 2008; Siler et al., 2019) and future conditions (Fyfe & etal, 2017; Gergel et al., 2017; Hamlet et al., 2005; Kapnick & Delworth, 2013; Klos et al., 2014; Rhoades, Ullrich, & Zarzycki, 2018), but there is a relative paucity of information on how the interannual variability of snowpack amount and timing might shift as the climate changes.

Changes in interannual variability of snowpack amount and timing would impact ecological, socioeconomic, and coupled social-ecological systems that rely on snow cover and melt, although these impacts are not as well established as the impact of changes in mean conditions. For example, the magnitude of interannual variability affects the reliability of reservoir inflows (Rhoades, Jones, & Ullrich, 2018), hydroelectric power generation (Fleming & Weber, 2012), and tourism (Scott et al., 2008). For each of these cases, low interannual variability may be associated with greater reliability, whereas high variability may increase the potential for

high snowfall and runoff years to offset the negative consequences of drought years. High interannual precipitation variability has been associated with reduced groundwater depletion because occasional high precipitation years can break a positive feedback between groundwater pumping and reservoir depletion (Apurv et al., 2017), though the contribution of variability in snow-related patterns, processes, and fluxes to surface and groundwater withdrawals is not as well documented.

One dimension of variability that may be particularly important is the degree of change between consecutive years. Precipitation whiplash—the occurrence of an extremely dry winter immediately followed by or preceding an extremely wet winter—is projected to increase in California (Swain et al., 2018), though snow-fall patterns will not reflect those of overall precipitation due to increasing temperatures. Recent multiyear snow droughts in the western United States, whether caused by unusually warm winters and/or low winter precipitation, have drawn attention to the causes and impacts of chronic snow droughts (Cooper et al., 2016; Hatchett & McEvoy, 2018; Ullrich et al., 2018). While studies of snow droughts have predominantly addressed snowpack amounts, consecutive years with early snow accumulation and ablation may also impact water resources (Jefferson et al., 2008). Current flood operations are guided by static rule curves that require reservoir drawdowns during fall months and neglect antecedent moisture conditions beyond the current season (Willis et al., 2011). The combination of required drawdowns and the potential for multiyear snow drought is a widespread threat to water availability from managed reservoirs.

Snowpack variability also affects ecological processes. For example, snow is important for threatened wildlife, such as wolverine (Copeland et al., 2010), and vegetation dynamics, such as timing of forest greenness (Trujillo et al., 2012). Earlier snow melt timing advances peak soil moisture timing (Harpold & Molotch, 2015) and flowering plant phenology (Dunne et al., 2003), increases vegetation water stress (Harpold et al., 2015), and is associated with increased wildfire activity (Westerling, 2016). While the importance of interannual variability and consecutive years with early snowmelt timing has not been formally established in this context, they may exacerbate stress on vegetation or affect plant community composition and productivity.

Previous studies have examined projected changes in interannual variability of temperature, precipitation, and snowpack across portions of the western United States. For example, in the Columbia River Basin, interannual temperature variability is projected to increase during summer and decrease in winter (Rupp et al., 2016). Interannual precipitation variability in the western United States is projected to increase, especially toward the end of the 21st century (Berg & Hall, 2015; Swain et al., 2018). Snowfall accumulation variability is projected to decrease in warmer-maritime regions and increase in colder continental regions (Lute et al., 2015). However, spatially explicit assessments of changes in the interannual variability of snowpack magnitude and timing are limited. These changes in snowpack variability may vary spatially in both direction and magnitude over relatively fine scales.

In this study, we assess projected changes in interannual variability of snowpack magnitude and timing, measured as annual maximum snow water equivalent (SWE<sub>max</sub>) and date of SWE<sub>max</sub> (DMS), across the western United States. These variables are selected to characterize the total amount of snow available to contribute to spring-summer runoff (SWE<sub>max</sub>) and the timing of the snow accumulation season (DMS). We also assess the frequency of consecutive years with very early or low SWE<sub>max</sub> occurring before or below the historical 25th percentile. Finally, we conduct a spatially explicit assessment of the frequency with which DMS occurred in specific months. This study is the first to assess how the magnitude and direction of change in variability are expected to vary spatially and differ between SWE<sub>max</sub> and DMS across the western United States. These findings provide important information for improving assessments of climate change impacts on water resources for socioeconomic, ecological, and coupled social-ecological systems.

# 2. Methods

Daily SWE data for the western United States were obtained from the Variable Infiltration Capacity (VIC) model (Liang et al., 1994) forced with downscaled climate model outputs. Ten global climate models (GCMs) from the Fifth Climate Model Intercomparison Project (CMIP5; Taylor et al., 2011) were selected based on their ability to credibly simulate temperature and precipitation patterns and variability across the northwestern United States (Abatzoglou & Rupp, 2017; Rupp et al., 2013). Many of these models also

performed well in California (Pierce et al., 2018). While historical performance may not predict future accuracy, culling of models is often used to guide impact-based modeling (e.g., Vano et al., 2015). Projected changes in seasonal temperature and precipitation for the subset of 10 models were largely consistent with a broader sample of GCM outputs for the region.

Data from each GCM was acquired for historical (1950–2005) and high-emissions representative concentration pathway (RCP) 8.5 (2006–2099) climate forcings. Daily GCM outputs were statistically downscaled to (1/16th)° resolution using multivariate adaptive constructed analogs (Abatzoglou & Brown, 2012) with the training data set of Livneh et al. (2013) and used to drive VIC. Comparisons of VIC simulations with observed SWE from the SNOTEL network showed good agreement, particularly with respect to interannual variability (Gergel et al., 2017) and have been widely used in hydroclimate research (e.g., Li et al., 2017). Study area maps are provided in the supporting information (Figures S1 and S2).

For each GCM, water year, and grid cell, annual SWE<sub>max</sub> and DMS were calculated from daily SWE for the water year beginning on 1 October for historical (1970–1999) and future (2050–2079) periods. When multiple days had the same value of SWE<sub>max</sub>, the last occurrence was recorded as DMS. The data were subset to grid cells where historical mean SWE<sub>max</sub> was greater than 100 mm; all subsequent calculations were conducted using this spatial domain. The interquartile range (IQR; 75th minus 25th percentile) of SWE<sub>max</sub> and DMS was calculated over the 30-year periods for the historical and future time periods for each GCM and grid cell. Mean values of SWE<sub>max</sub> and DMS are presented in the supporting information (Figures S3–S5).

We calculated changes in the frequency of consecutive low SWE snow drought, defined as years for which both the current and antecedent year had SWE<sub>max</sub> below the historic 25th percentile. Similarly, we calculated changes in the frequency of consecutive early SWE snow drought, defined as multiple years with DMS before the historic 25th percentile. In cases where three consecutive years had SWE<sub>max</sub> below the first quartile, two years would be tallied. Sensitivity analyses with 4-year durations were conducted, though we urge caution in interpreting GCM results pertaining to the ability of GCMs to capture lower-frequency climate variability (Abatzoglou & Rupp, 2017).

We used a bootstrap approach to test statistical significance of changes in variability. For each GCM and grid cell, we resampled 30 years with replacement 100 times from the historical model years 1970-1999 and calculated variability metrics for each sample. Differences were deemed significant where the variability calculated for model years 2050–2079 fell outside of the historical 5th–95th percentiles. As GCM variability has been cited as a key source of uncertainty for early-century to midcentury regional climate projections (e.g., Chegwidden et al., 2019; Chen et al., 2011; Hawkins & Sutton, 2011), we consider changes to be robust when significant changes in the same direction are observed in at least five of 10 GCMs. Results are also presented as supporting information in an interactive tool at https://snowvariability.nkn.uidaho.edu/.

# 3. Results and Discussion

## 3.1. SWE<sub>max</sub> Variability

Historical SWE<sub>max</sub> IQR is largest in high elevation, cold regions of the Sierra Nevada and Cascades, and lower across the colder interior mountains. Lower elevations throughout the study area exhibit lower IQR due to lower upper quartile values of SWE<sub>max</sub> (Figure 1a). Changes in SWE<sub>max</sub> IQR from historical to future periods show distinct spatial patterns (Figure 1b). In lower elevations of maritime mountains, SWE<sub>max</sub> IQR decreases due to greater declines in the upper versus bottom quartile of SWE<sub>max</sub> distributions, suggesting that there will be fewer years with what would historically be considered an above average snowpack. In higher elevations of the Sierra Nevada, Cascades, Northern Rockies, and Idaho Batholith SWE<sub>max</sub> IQR increases. In the southern Rockies, both quartiles decreased (Figure S7), such that SWE<sub>max</sub> IQR was relatively unchanged. Across the entire domain, the largest decreases in SWE<sub>max</sub> IQR occur at sites where historical average winter (November–March) temperatures are greater than 0 °C (Figure S7) and SWE<sub>max</sub> IQR are not well explained by temperature. Historical winter precipitation and changes in both winter precipitation and temperature IQR were generally not strongly linked to changes in SWE<sub>max</sub> IQR (Figures S8–S11).

At least 5 out of 10 GCMs simulate significant decreases in SWE<sub>max</sub> IQR for 21.8% of grid cells, while 1.0% of grid cells meet this criterion for increases (Figures S13 and S14). Sites with significant decreases in SWE<sub>max</sub>





**Figure 1.** (a) Historical and (b) change in SWE<sub>max</sub> IQR. (c) Historical and (d) change in percent of water years classified as 2-year consecutive low SWE<sub>max</sub> snow droughts. SWE<sub>max</sub> = annual maximum snow water equivalent; IQR = interquartile range.

IQR are predominantly in warmer regions that are likely to experience a transition toward a greater fraction of precipitation falling as rain, rather than snow (Klos et al., 2014). 76.7% (4.4%) of pixels show a robust increase (decrease) in consecutive snow droughts.

Historically, the frequency of consecutive low SWE years showed minimal spatial variability with an average of 6.6% of water years identified as part of a 2-year or longer snow drought (Figure 1c). In 2050–2079, an average of 42.2% of water years classify as consecutive snow droughts. These changes are greatest in maritime regions and across the large area that comprises the lower elevations of the northern Rockies (Figure 1d). Spatial patterns of change in consecutive low SWE years are broadly similar to percentage changes in mean SWE<sub>max</sub> (Figure S5). The average frequency of a 4-year consecutive low snow drought increased from 0.26% of water years to 25.0%.

To illustrate the spatially complex nature of changes in variability,  $SWE_{max}$  IQR is depicted for three grid cells along a transect in the Sierra Nevada for a single GCM (Figure 2; other GCMs in Figures S16 and S17). Distributions of annual  $SWE_{max}$  for the historical and mid-21st century cases reveal distinctly different patterns of change across the transect. At the low- and midelevation pixels, which were historically near the winter 0 °C isotherm, zero or near-zero  $SWE_{max}$  values become increasingly common in the future, and both





**Figure 2.** (a, c) Maps of changes in SWE<sub>max</sub> and DMS IQR for CNRM-CM5 with transects marked at 38.0°N (Figure S15 for area map). (b, d) Distributions of (b) SWE<sub>max</sub> and (d) DMS in CNRM-CM5 for three points on the transect marked in (a) and (c). Vertical lines indicate first and third quartiles in the historical and mid-21st century cases. Historical average winter temperature and elevation are noted. SWE<sub>max</sub> = annual maximum snow water equivalent; IQR = interquartile range; DMS = date of SWE<sub>max</sub>.

the upper and lower quartile values decrease (Figure 2b). At the low elevation pixel, the upper quartile decreases much more than the lower quartile, decreasing the  $SWE_{max}$  IQR by over 60%. At the midelevation pixel, the lower quartile decreases more than the upper, doubling  $SWE_{max}$  IQR. Finally, at the highest elevation pixel, the upper quartile increases, likely due to increasing winter precipitation (Rupp et al., 2017), and the lower quartile decreases, likely due to warmer years with low  $SWE_{max}$ , producing a 130% increase in  $SWE_{max}$  IQR.

These findings suggest that  $SWE_{max}$  IQR decreases and consecutive low SWE years increase in areas that are near the historical 0 °C isotherm, where warming causes a shift from snow to rain (Klos et al., 2014), primarily due to the loss of years with exceptionally deep snowpack. This is in agreement with Lute et al. (2015), who found a maritime-continental gradient of changes in  $SWE_{max}$  standard deviation, with increases in colder continental inland ranges and decreases in warmer maritime regions. The continental-scale patterns of historical  $SWE_{max}$  IQR and large decreases in  $SWE_{max}$  IQR in maritime regions are likely due to the contribution of snowfall intensity and extreme events to interannual variability (Lute & Abatzoglou, 2014). In maritime regions, larger  $SWE_{max}$  years depend on a few large events (Guan et al., 2010), which are susceptible to warming and precipitation phase shifts from snow to rain (Lute et al., 2015) and increased winter ablation (Kapnick & Hall, 2012). The warming-induced loss of a few large snowfall events in years that would otherwise have large  $SWE_{max}$  values would reduce  $SWE_{max}$ , producing a large decline in IQR.

### 3.2. DMS variability

Historically, DMS IQR was largest at lower elevations in the Sierra Nevada and Cascades and lower in the colder Rocky Mountains (Figure 3a). This pattern illustrates that peak snowpack timing was historically most variable in warmer regions with high interannual precipitation variability and relatively intermittent





Figure 3. (a) Historical and (b) change in DMS IQR. (c) Historical and (d) change in frequency of consecutive early SWE years. DMS = date of SWE<sub>max</sub>; SWE = snow water equivalent; IQR = interquartile range.

snowpack. Conversely, changes in DMS IQR display complex spatial patterns (Figure 3b) that are not well explained by historical climate or changes in climatic variability (Figures S8–S11). DMS IQR generally decreases in the highest elevation, coldest regions and increases in warmer areas, though there are some exceptions to this pattern, such as the foothills of the Oregon Cascades. More areas exhibit significant increases in DMS IQR that are robust across GCMs (24.8% of pixels) than decreases (0.6%; Figure S13).

The complex spatial patterns of change in DMS variability is illustrated through inspection of grid cells along a transect (Figure 2d). In the lowest elevation Sierra Nevada pixel, DMS was historically quite variable, but the earliest quartile advances more than the latest, so DMS IQR increases from 34 to 57 days in the future case. In the midelevation pixel, the first and third quartiles change by the same amount, and variability is unchanged. At the highest elevation, the latest quartile of years changes much more than the earliest quartile, and DMS IQR decreases from 31 to 19 days.

The increase in DMS IQR in warmer regions is likely indicative of sites in the snow-to-rain transition zone that historically had relatively low DMS IQR. As temperatures warm and snowpack declines, DMS becomes increasingly dependent on the synchrony of precipitation events and subfreezing temperatures, and thus more variable. For example, in the Sierra Nevada and Cascades, a few large storms deliver a large fraction



# **Geophysical Research Letters**



Figure 4. (a) Historical and mid-21st century frequency of date of peak SWE occurring in each month. (b) Change in percent of pixels from historical (left) to mid-21st century (right) for which DMS occurs in a given month the majority of the time, with 10-GCM mean in black. SWE = snow water equivalent; global; GCM = global climate model; DMS = date of SWE<sub>max</sub>.

of the annual snowfall (Lute & Abatzoglou, 2014), which may be of heightened importance in a warmer climate with fewer days conducive to snowfall (e.g., Lute et al., 2015). Late DMS years can be heavily affected when one early winter storm produces rain, rather than snow. To the extent that DMS is related to runoff timing, increasing variability suggests increased runoff timing variability, though these impacts will be mediated by post-DMS ablation rates and runoff generation processes, which may in turn be affected by climate change (e.g., Barnhart et al., 2016; Musselman et al., 2017).

Historically, an average of 6.4% of water years were consecutive early SWE years; this number increases to 56.7% in the mid-21st century case. GCM agreement was high, with 96.9% of grid cells meeting our criteria for robust increase in consecutive early SWE years. As with SWE<sub>max</sub>, the historical frequency of consecutive early DMS years does not show obvious spatial patterns (Figure 3c). Consecutive early SWE years increase across the domain, with a pattern of change that is similar to change in mean DMS (Figure S4), with greatest changes in the northern Rockies and Cascades, but without the maritime-to-continental climate gradient evident for changes in SWE<sub>max</sub>. For 4-year durations, the average frequency of consecutive early DMS years increases from 0.27% to 38.0%.

A frequency analysis of the historical and potential future timing of DMS summarizes this variability (Figure 4). We define "reliable DMS" as cases where a grid cell has at least 50% of DMS values in a given

month. April was the predominant month in which  $SWE_{max}$  occurred historically (29.0% of grid cells had reliable April DMS). March DMS was more common at lower elevations, and May was relatively common at higher elevations, particularly in the continental interior. In 2050–2079, April is no longer the most common month in which peak SWE occurs, with only 15.5% of grid cells having reliable April DMS. Pixels with reliable DMS in May decrease from 9.9 to 1.2% of grid cells. Instead, peak SWE values in March and February become increasingly common. These findings are broadly consistent with existing literature showing that DMS has shifted earlier and is projected to continue to do so in future climates (Kapnick & Hall, 2010; Montoya et al., 2014), though here we add more spatially explicit and detailed projections. Moreover, these changes reflect increasing variability of DMS, as the total percentage of pixels that had no months with reliable DMS increased from 37.1 to 51.1%.

Model agreement on significant changes in SWE<sub>max</sub> and DMS IQR varies regionally. On average, model agreement on changes in SWE<sub>max</sub> and DMS IQR is lower than model agreement on changes in means (Figure S13). Warming is a robust feature of modeled future climates, while changes in precipitation and temperature variability exhibit greater uncertainty and model disagreement (Rupp et al., 2016, 2017). Snowpack variability may be affected by warming, changes in precipitation magnitude, and spatiotemporal variability of temperature and precipitation, as well as other contributors to the snowcover energy balance, such as shortwave radiation (Musselman et al., 2017; Painter et al., 2017) and atmospheric humidity (Harpold & Brooks, 2018). Different snow models may affect results but have previously been identified as a relatively small source of uncertainty (Chen et al., 2011). Further work may be needed to assess the robustness of these results given the multiple sources of uncertainty, including climate forcing due to intermodel, interscenario, and internal variability (Hawkins & Sutton, 2011), choice of downscaling approach and reference observational data (Alder & Hostetler, 2018), and choice of hydrologic model. To the extent that GCMs agree on changes in snowpack variability, we propose that these changes are likely incurred due to warming, but future work should quantitatively assess physical mechanisms for changes in snowpack variability.

# 4. Implications and Conclusions

Interannual variability of  $SWE_{max}$  in the western United States is projected to change, with large decreases in IQR for regions transitioning from snow- to rain-dominated climates, particularly in maritime regions, and smaller changes in cooler continental climates. In contrast, DMS may become more variable across much of the western United States. Spatial patterns of the magnitude and direction of these trends are critical for understanding their impacts. Here, we discuss several potential implications of these findings.

For water resources operations, regions with increases in interannual variability of runoff volume and timing that have large engineered or natural storage may be more resilient to changes than those with less storage, particularly when storage exceeds the average annual discharge (Gaupp et al., 2015; Langbein, 1959). The impacts of snowpack magnitude and timing variability on water resources also depend on the combined effects of snow and rain on runoff, particularly as previously snowmelt-dominated systems experience increasing contributions from rainfall (e.g., Knowles et al., 2006; Kormos et al., 2016). Changing snowpack variability will likely be very different from changing precipitation variability, which is generally projected to increase (e.g., Konapala et al., 2017; Pendergrass et al., 2017; Swain et al., 2018), and the combinations of these changes will determine changes in water resources. The increased frequency of consecutive low SWE years will also affect water resources, requiring improved early drought detection methods (AghaKouchak et al., 2015) and optimization of reservoir operation rule curves to account for antecedent storage (Anderson et al., 2008; Ralph et al., 2014; Willis et al., 2011) as the natural snowpack storage reservoir is depleted.

Recreational activities, such as ski resort operations, which depend on a minimum amount and relatively early snowpack accumulation as well as reliability of snow conditions coinciding with peak visitation periods (Scott et al., 2008), will likely be affected by altered interannual variability. Our results suggest that low snowpack years will be more common, with reduced interannual variability and that the frequency of consecutive low SWE<sub>max</sub> years will increase for ski resorts at lower elevations where precipitation will increasingly fall as rain rather than snow. Similar findings apply for ecosystem functions that are influenced by interannual snowpack variability. For example, deep snowpack facilitates subalpine seedling establishment (Andrus et al., 2018). Reduced variability of SWE<sub>max</sub> and the loss of high SWE<sub>max</sub> years could limit

seedling establishment and alter successional dynamics. High snowpack years also limit early season fire activity in many mountainous regions of the West (Abatzoglou & Kolden, 2013; Westerling, 2016); loss of these years could enable more consistent early onset of fire activity in flammability-limited regions, barring increased spring and early-summer rainfall.

While studies of the importance of average snowpack conditions for water resources and ecosystems abound, the impacts of changing variability on these systems are less well established. Our results suggest that snowpack variability will be substantially altered in the future climates considered here, with robust increases in the frequency of recurrent snow drought and reduced interannual variability of annual SWE<sub>max</sub>. To the extent that changes in snowpack variability affect water resources and ecosystem function, climate change impact studies and adaptation planning efforts should account for future changes in snowpack variability.

#### References

Abatzoglou, J. T., & Brown, T. J. (2012). A comparison of statistical downscaling methods suited for wildfire applications. *International Journal of Climatology*, 32(5), 772–780. https://doi.org/10.1002/joc.2312

Abatzoglou, J. T., & Kolden, C. A. (2013). Relationships between climate and macroscale area burned in the western United States. International Journal of Wildland Fire, 22(7), 1003. https://doi.org/10.1071/WF13019

AghaKouchak, A., Farahmand, A., Melton, F. S., Teixeira, J., Anderson, M. C., Wardlow, B. D., & Hain, C. R. (2015). Remote sensing of drought: Progress, challenges and opportunities. *Reviews of Geophysics*, 53, 452–480. https://doi.org/10.1002/2014RG000456

Alder, J. R., & Hostetler, S. W. (2018). The dependence of hydroclimate projections in snow-dominated regions of the western U.S. on the choice of statistically downscaled climate data. *Water Resources Research*, 55, 2279–2300. https://doi.org/10.1029/2018WR023458 Anderson, J., Chung, F., Anderson, M., Brekke, L., Easton, D., Ejeta, M., et al. (2008). Progress on incorporating climate change into

management of California's water resources. *Climatic Change*, 87(S1), 91–108. https://doi.org/10.1007/s10584-007-9353-1

Andrus, R. A., Harvey, B. J., Rodman, K. C., Hart, S. J., & Veblen, T. T. (2018). Moisture availability limits subalpine tree establishment. *Ecology*, 99(3), 567–575. https://doi.org/10.1002/ecy.2134

Apurv, T., Sivapalan, M., & Cai, X. (2017). Understanding the role of climate characteristics in drought propagation. *Water Resources Research*, *53*, 9304–9329. https://doi.org/10.1002/2017WR021445

- Barnett, T. P., Pierce, D. W., Hidalgo, H. G., Bonfils, C., Santer, B. D., Das, T., et al. (2008). Human-induced changes in the hydrology of the Western United States. Science, 319(5866), 1080–1083. https://doi.org/10.1126/science.1152538
- Barnhart, T. B., Molotch, N. P., Livneh, B., Harpold, A. A., Knowles, J. F., & Schneider, D. (2016). Snowmelt rate dictates streamflow. Geophysical Research Letters, 43, 8006–8016. https://doi.org/10.1002/2016GL069690
- Berg, N., & Hall, A. (2015). Increased interannual precipitation extremes over California under climate change. *Journal of Climate*, 28(16), 6324–6334. https://doi.org/10.1175/JCLI-D-14-00624.1
- Chegwidden, O. S., Nijssen, B., Rupp, D. E., Arnold, J. R., Clark, M. P., Hamman, J. J., et al. (2019). How do modeling decisions affect the spread among hydrologic climate change projections? Exploring a large ensemble of simulations across a diversity of hydroclimates. *Earth's Future*. https://doi.org/10.1029/2018EF001047
- Chen, J., Brissette, F. P., Poulin, A., & Leconte, R. (2011). Overall uncertainty study of the hydrological impacts of climate change for a Canadian watershed. *Water Resources Research*, 47, W12509. https://doi.org/10.1029/2011WR010602
- Cooper, M. G., Nolin, A. W., & Safeeq, M. (2016). Testing the recent snow drought as an analog for climate warming sensitivity of Cascades snowpacks. Environmental Research Letters, 11(8), 084009. https://doi.org/10.1088/1748-9326/11/8/084009
- Copeland, J. P., McKelvey, K. S., Aubry, K. B., Landa, A., Persson, J., Inman, R. M., et al. (2010). The bioclimatic envelope of the wolverine (Gulo gulo): do climatic constraints limit its geographic distribution? *Canadian Journal of Zoology*, 88(3), 233–246. https://doi.org/ 10.1139/Z09-136
- Dunne, J. A., Harte, J., & Taylor, K. J. (2003). Subalpine meadow flowering phenology responses to climate change: Integrating experimental and gradient methods. *Ecological Monographs*, 73(1), 69–86. https://doi.org/10.1890/0012-9615(2003)073[0069: SMFPRT]2.0.CO;2
- Fleming, S. W., & Weber, F. A. (2012). Detection of long-term change in hydroelectric reservoir inflows: Bridging theory and practise. Journal of Hydrology, 470–471, 36–54. https://doi.org/10.1016/j.jhydrol.2012.08.008
- Fyfe, J. C., et al. (2017). Large near-term projected snowpack loss over the western United States. *Nature Communications*, 8. https://doi. org/10.1038/ncomms14996
- Gaupp, F., Hall, J., & Dadson, S. (2015). The role of storage capacity in coping with intra- and inter-annual water variability in large river basins. Environmental Research Letters, 10(12), 125001. https://doi.org/10.1088/1748-9326/10/12/125001
- Gergel, D. R., Nijssen, B., Abatzoglou, J. T., Lettenmaier, D. P., & Stumbaugh, M. R. (2017). Effects of climate change on snowpack and fire potential in the western USA. Climatic Change, 141(2), 287–299. https://doi.org/10.1007/s10584-017-1899-y

Guan, B., Molotch, N. P., Waliser, D. E., Fetzer, E. J., & Neiman, P. J. (2010). Extreme snowfall events linked to atmospheric rivers and surface air temperature via satellite measurements. *Geophysical Research Letters*, 37, L20401. https://doi.org/10.1029/2010GL044696

- Hamlet, A. F., & Lettenmaier, D. P. (2007). Effects of 20th century warming and climate variability on flood risk in the western U.S. Water Resources Research, 43, W06427. https://doi.org/10.1029/2006WR005099
- Hamlet, A. F., Mote, P. W., Clark, M. P., & Lettenmaier, D. P. (2005). Effects of temperature and precipitation variability on snowpack trends in the Western United States. *Journal of Climate*, 18(21), 4545–4561. https://doi.org/10.1175/JCLI3538.1
- Harpold, A. A., & Brooks, P. D. (2018). Humidity determines snowpack ablation under a warming climate. *Proceedings of the National Academy of Sciences*, 115(6), 1215–1220. https://doi.org/10.1073/pnas.1716789115
- Harpold, A. A., & Molotch, N. P. (2015). Sensitivity of soil water availability to changing snowmelt timing in the western U.S. Geophysical Research Letters, 42, 8011–8020. https://doi.org/10.1002/2015GL065855

#### Acknowledgments

This work was supported by NSF IGERT Program Grant 1249400 (A. M. and T. L.) and NOAA Grant NA150AR4310145 (J. A.) Data are available at http://climate.nkn.uidaho. edu/IntegratedScenarios/. Comments from Valeriy Ivanov, Matt Cooper, and three anonymous reviewers greatly improved this manuscript.

Abatzoglou, J. T., & Rupp, D. E. (2017). Evaluating climate model simulations of drought for the northwestern United States. International Journal of Climatology, 37(S1), 910–920. https://doi.org/10.1002/joc.5046

Harpold, A. A., Molotch, N. P., Musselman, K. N., Bales, R. C., Kirchner, P. B., Litvak, M., & Brooks, P. D. (2015). Soil moisture response to snowmelt timing in mixed-conifer subalpine forests. *Hydrological Processes*, 29(12), 2782–2798. https://doi.org/10.1002/hyp.10400

Hatchett, B. J., & McEvoy, D. J. (2018). Exploring the origins of snow drought in the Northern Sierra Nevada, California. Earth Interactions, 22(2), 1–13. https://doi.org/10.1175/EI-D-17-0027.1

Hawkins, E., & Sutton, R. (2011). The potential to narrow uncertainty in projections of regional precipitation change. *Climate Dynamics*, 37(1), 407–418. https://doi.org/10.1007/s00382-010-0810-6

Jefferson, A., Nolin, A., Lewis, S., & Tague, C. (2008). Hydrogeologic controls on streamflow sensitivity to climate variation. *Hydrological Processes*, 22(22), 4371–4385. https://doi.org/10.1002/hyp.7041

Kapnick, S., & Hall, A. (2010). Observed climate-snowpack relationships in California and their implications for the future. Journal of Climate, 23(13), 3446–3456. https://doi.org/10.1175/2010JCLI2903.1

Kapnick, S., & Hall, A. (2012). Causes of recent changes in western North American snowpack. Climate Dynamics, 38(9–10), 1885–1899. https://doi.org/10.1007/s00382-011-1089-y

Kapnick, S. B., & Delworth, T. L. (2013). Controls of global snow under a changed climate. Journal of Climate, 26(15), 5537–5562. https:// doi.org/10.1175/JCLI-D-12-00528.1

Klos, P. Z., Link, T. E., & Abatzoglou, J. T. (2014). Extent of the rain-snow transition zone in the western U.S. under historic and projected climate. *Geophysical Research Letters*, 41, 4560–4568. https://doi.org/10.1002/2014GL060500

Knowles, N., Dettinger, M. D., & Cayan, D. R. (2006). Trends in snowfall versus rainfall in the Western United States. *Journal of Climate*, 19(18), 4545–4559. https://doi.org/10.1175/JCLI3850.1

Konapala, G., Mishra, A., & Leung, L. R. (2017). Changes in temporal variability of precipitation over land due to anthropogenic forcings. Environmental Research Letters, 12(2), 024009. https://doi.org/10.1088/1748-9326/aa568a

Kormos, P. R., Luce, C. H., Wenger, S. J., & Berghuijs, W. R. (2016). Trends and sensitivities of low streamflow extremes to discharge timing and magnitude in Pacific Northwest mountain streams. *Water Resources Research*, 52, 4990–5007. https://doi.org/10.1002/ 2015WR018125

Langbein, W. B. (1959). Water yield and reservoir storage in the United States. Washington, DC: U.S. Government Printing Office. https://pubs.usgs.gov/circ/1959/0409/report.pdf

Li, D., Wrzesien, M. L., Durand, M., Adam, J., & Lettenmaier, D. P. (2017). How much runoff originates as snow in the western United States, and how will that change in the future?: Western U.S. Snowmelt-Derived Runoff. *Geophysical Research Letters*, 44, 6163–6172. https://doi.org/10.1002/2017GL073551

Liang, X., Lettenmaier, D. P., Wood, E. F., & Burges, S. J. (1994). A simple hydrologically based model of land surface water and energy fluxes for general circulation models. *Journal of Geophysical Research*, 99(D7), 14,415–14,428. https://doi.org/10.1029/ 94JD00483

Livneh, B., Rosenberg, E. A., Lin, C., Nijssen, B., Mishra, V., Andreadis, K. M., et al. (2013). A long-term hydrologically based dataset of land surface fluxes and states for the conterminous United States: Update and extensions. *Journal of Climate*, 26(23), 9384–9392. https://doi. org/10.1175/JCLI-D-12-00508.1

Lute, A. C., & Abatzoglou, J. T. (2014). Role of extreme snowfall events in interannual variability of snowfall accumulation in the western United States. Water Resources Research, 50, 2874–2888. https://doi.org/10.1002/2013WR014465

Lute, A. C., Abatzoglou, J. T., & Hegewisch, K. C. (2015). Projected changes in snowfall extremes and interannual variability of snowfall in the western United States. Water Resources Research, 51, 960–972. https://doi.org/10.1002/2014WR016267

Montoya, E. L., Dozier, J., & Meiring, W. (2014). Biases of April 1 snow water equivalent records in the Sierra Nevada and their associations with large-scale climate indices. *Geophysical Research Letters*, 41, 5912–5918. https://doi.org/10.1002/2014GL060588

Mote, P. W. (2006). Climate-driven variability and trends in mountain snowpack in Western North America. Journal of Climate, 19(23), 6209–6220. https://doi.org/10.1175/JCLI3971.1

Mote, P. W., Li, S., Lettenmaier, D. P., Xiao, M., & Engel, R. (2018). Dramatic declines in snowpack in the western US. *Npj Climate and Atmospheric Science*, 1(1), 2. https://doi.org/10.1038/s41612-018-0012-1

Musselman, K. N., Clark, M. P., Liu, C., Ikeda, K., & Rasmussen, R. (2017). Slower snowmelt in a warmer world. *Nature Climate Change*, 7(3), 214–219. https://doi.org/10.1038/nclimate3225

Painter, T. H., Skiles, S. M., Deems, J. S., Brandt, W. T., & Dozier, J. (2017). Variation in rising limb of colorado river snowmelt runoff hydrograph controlled by dust radiative forcing in snow. *Geophysical Research Letters*, 45, 797–808. https://doi.org/10.1002/ 2017GL075826

Pendergrass, A. G., Knutti, R., Lehner, F., Deser, C., & Sanderson, B. M. (2017). Precipitation variability increases in a warmer climate. Scientific Reports, 7(1), 17966. https://doi.org/10.1038/s41598-017-17966-y

Pierce, D. W., Barnett, T. P., Hidalgo, H. G., Das, T., Bonfils, C., Santer, B. D., et al. (2008). Attribution of declining Western U.S. Snowpack to human effects. *Journal of Climate*, 21(23), 6425–6444. https://doi.org/10.1175/2008JCLI2405.1

- Pierce, D. W., Kalansky, J. F., & Cayan, D. R. (2018). Climate, drought, and sea level rise scenarios for the fourth California climate assessment. California's Fourth Climate Change Assessment, California Energy Commission, Publication Number: CNRA-CEC-2018-006.
- Ralph, F. M., Dettinger, M., White, A., Reynolds, D., Cayan, D., Schneider, T., et al. (2014). A Vision for future observations for Western U.S. Extreme precipitation and flooding. *Journal of Contemporary Water Research & Education*, 153(1), 16–32. https://doi.org/10.1111/j.1936-704X.2014.03176.x

Rhoades, A. M., Jones, A. D., & Ullrich, P. A. (2018). The changing character of the California Sierra Nevada as a natural reservoir. *Geophysical Research Letters*, 45, 13,008–13,019. https://doi.org/10.1029/2018GL080308

Rhoades, A. M., Ullrich, P. A., & Zarzycki, C. M. (2018). Projecting 21st century snowpack trends in western USA mountains using variableresolution CESM. *Climate Dynamics*, 50(1–2), 261–288. https://doi.org/10.1007/s00382-017-3606-0

Rupp, D. E., Abatzoglou, J. T., Hegewisch, K. C., & Mote, P. W. (2013). Evaluation of CMIP5 20th century climate simulations for the Pacific Northwest USA. Journal of Geophysical Research: Atmospheres, 118, 10,884–10,906. https://doi.org/10.1002/jgrd.50843

Rupp, D. E., Abatzoglou, J. T., & Mote, P. W. (2016). Projections of 21st century climate of the Columbia River Basin. Climate Dynamics, 49(5-6), 1783–1799. https://doi.org/10.1007/s00382-016-3418-7

Rupp, D. E., Li, S., Mote, P. W., Shell, K. M., Massey, N., Sparrow, S. N., et al. (2017). Seasonal spatial patterns of projected anthropogenic warming in complex terrain: A modeling study of the western US. *Climate Dynamics*, 48(7–8), 2191–2213. https://doi.org/10.1007/ s00382-016-3200-x

Scott, D., Dawson, J., & Jones, B. (2008). Climate change vulnerability of the US Northeast winter recreation—Tourism sector. Mitigation and Adaptation Strategies for Global Change, 13(5–6), 577–596. https://doi.org/10.1007/s11027-007-9136-z

- Siler, N., Proistosescu, C., & Po-Chedley, S. (2019). Natural variability has slowed the decline in Western U.S. Snowpack since the 1980s. Geophysical Research Letters, 46, 346–355. https://doi.org/10.1029/2018GL081080
- Swain, D. L., Langenbrunner, B., Neelin, J. D., & Hall, A. (2018). Increasing precipitation volatility in twenty-first-century California. *Nature Climate Change*, 8(5), 427–433. https://doi.org/10.1038/s41558-018-0140-y
- Taylor, K. E., Stouffer, R. J., & Meehl, G. A. (2011). An overview of CMIP5 and the experiment design. Bulletin of the American Meteorological Society, 93(4), 485–498. https://doi.org/10.1175/BAMS-D-11-00094.1
- Trujillo, E., Molotch, N. P., Goulden, M. L., Kelly, A. E., & Bales, R. C. (2012). Elevation-dependent influence of snow accumulation on forest greening. *Nature Geoscience*, 5(10), 705–709. https://doi.org/10.1038/ngeo1571
- Ullrich, P. A., Xu, Z., Rhoades, A. M., Dettinger, M. D., Mount, J. F., Jones, A. D., & Vahmani, P. (2018). California's drought of the future: A midcentury recreation of the exceptional conditions of 2012–2017. *Earth's Future*, 6(11), 1568–1587. https://doi.org/10.1029/ 2018EF001007
- Vano, J. A., Kim, J. B., Rupp, D. E., & Mote, P. W. (2015). Selecting climate change scenarios using impact-relevant sensitivities. *Geophysical Research Letters*, 42, 5516–5525. https://doi.org/10.1002/2015GL063208
- Westerling, A. L. (2016). Increasing western US forest wildfire activity: Sensitivity to changes in the timing of spring. *Philosophical Transactions of the Royal Society B: Biological Sciences*, 371(1696). https://doi.org/10.1098/rstb.2015.0178
- Willis, A. D., Lund, J. R., Townsley, E. S., & Faber, B. A. (2011). Climate change and flood operations in the Sacramento Basin, California. San Francisco Estuary and Watershed Science.